Author's response to reviews

Title: Association between Socioeconomic Status and Pain, Function and Pain Catastrophizing at Presentation for Total Knee Arthroplasty

Authors:

Candace H Feldman (candace.feldman@gmail.com)
Yan Dong (yad509@mail.harvard.edu)
Jeffrey N Katz (jinkatz@partners.org)
Laurel A. Donnell-Fink (ldonnell-fink@partners.org)
Elena Losina (elosina@partners.org)

Version: 2
Date: 30 September 2014

Author's response to reviews:

Line numbers have been added, section titles have been changed, and the information required has been listed under an acknowledgements section.